Author's response to reviews

Title: Common Polymorphisms of Calpain-10 and the Risk of Type 2 Diabetes in a Tunisian Arab Population: a case-control study

Authors:

Intissar Ezzidi (intissar_z@yahoo.fr)
Safia Messaoudi (massoudisafia@yahoo.fr)
Molka Chaieb (molkachadli@yahoo.fr)
Maha Kacem (kacem_maha@yahoo.fr)
Ghada M Al-Khateeb (ghadaalkhateeb@yahoo.com)
Touhami Mahjoub (mahjoubtouhami@yahoo.fr)
Wassim Y Almawi (wassim@agu.edu.bh)
Nabil Mtiraoui (mtiraouinabil@yahoo.fr)

Version: 7 Date: 11 May 2010

Author's response to reviews: see over
May 11, 2010

Scott Edmunds, Ph.D.
Senior Scientific Editor
BMC-series Journals
BioMed Central
Floor 6, 236 Gray's Inn Road
London, WC1X 8HL

Dear Dr. Edmunds,

Please find enclosed a revised version of the manuscript entitled:

Common Polymorphisms of Calpain-10 and the Risk of Type 2 Diabetes in a Tunisian Arab Population: a case-control study

Intissar EZZIDI, Amira Turki, Safia MESSAOUDI, Molka CHAIEB, Maha KACEM, Ghada M Al-KHATEEB, Touhami MAHJOUB, Wassim Y ALMAWI and Nabil MTIRAOUl, which we submit for consideration for publication in BMC Medical Genetics, after having addressed formatting changes.

None of the authors has any potential financial conflict of interest related to this manuscript.

All authors have carefully examined and approved of the changes made to the manuscript, and understand and accept that in the event of its publication, all copyright shall be transferred to BMC Medical Genetics. We look forward to hear from you favorably in due course.

Sincerely,

Nabil MTIRAOUl, Ph.D.
Biologie et Génétique des Maladies Hématologiques et Auto-immunes
Faculté de Pharmacie de Monastir
Université de Monastir
Monastir, PO BOX 5000,
Tunisie

Tel: +216 96 27 24 81
Fax: +216 73 46 18 30

Email: mtiraouinabil@yahoo.fr